Author's response to reviews

Title: Determinants of infant growth in Eastern Uganda: a community-based cross-sectional study

Authors:

Ingunn Marie Stadskleiv Engebretsen (ingunn.engebretsen@cih.uib.no)
Thorkild Tylleskar (thorkild.tylleskar@cih.uib.no)
Henry Wamani (wamanih@yahoo.com.ac.ug)
Charles Karamagi (ckaramagi2000@yahoo.com)
James K Tumwine (jatumwine@imul.com)

Version: 5 Date: 11 December 2008

Author's response to reviews: see over
To
The Editor
BMC Public Health

Dear Editor,

Re: Re-submission of manuscript: “Determinants of infant growth in Eastern Uganda: a community-based cross-sectional study”

We are pleased to re-submit a final version of our manuscript by Ingunn Marie S Engebretsen, Thorkild Tylleskär, Henry Wamani, Charles Karamagi and James Tumwine entitled “Determinants of infant growth in Eastern Uganda: a community-based cross-sectional study.”

We have on the basis of the BMC Public Health template and the e-mail received from December 8th 2008 revised the document according to the formatting changes required.

Thank you for the editorial process and guidance. Please let me know if there are further issues.

Yours sincerely,

Ingunn Marie S Engebretsen, MD
Correspondence:
  Ingunn Marie S Engebretsen
  Centre for International Health, CIH
  Armauer Hansen Building
  NO – 5021 Bergen, Norway

E-mail: ingunn.engebretsen@cih.uib.no